

Author's response to reviews

Title: Demographic differences in and correlates of perceived body image discrepancy among urban adolescent girls: a cross-sectional study

Authors:

Lorraine Robbins (robbin76@msu.edu)

Jiying Ling (Jiying.Ling@hc.msu.edu)

Kenneth Resnicow (kresnic@umich.edu)

Version: 2 Date: 18 Sep 2017

Author's response to reviews:

Reviewer 1: This is a really important paper that complements the existing knowledge about the relationship between body image and physical activity by adding data related to body size discrepancy, and objective measures of physical activity. I am satisfied with the modifications that have been made to the paper. My only suggestion would be that the authors add clarity around the procedure in terms of when the CV testing was conducted by using the terminology of the test- I had to read it a few times to figure out the order of the activities/measures.

Response: Thank you for stating that this paper is important.

Thank you for stating that you are satisfied with the modifications that have been made to the paper. We are sorry for the lack of clarity in the one area. We added additional PACER information in the procedure and measures sections (pages 10 and 11). We added when the PACER was conducted to the procedure. We hope that the info is now clear.

Thank you for your comment.